Author's response to reviews

Title: Collaborative Treatment of Huge Intrathoracic Meningoceles Associated with Neurofibromatosis Type 1: a case report

Authors:

Deog Gon Cho (ebstein8@daum.net)
Yong Jin Chang (tsmate@hanmail.net)
Kyu Do Cho (kyudias@catholic.ac.kr)
Jae Taek Hong (jatagi15@paran.com)

Version: 4 Date: 14 May 2015

Author's response to reviews: see over
Subject: Submission of a case report for evaluation

Date: 11 May 2015

Dear Editor

I am enclosing here with a manuscript entitled “Collaborative Treatment of Huge Intrathoracic Meningoceles Associated with Neurofibromatosis Type 1” for publication in Interactive CardioVascular and Thoracic Surgery for possible evaluation. With the submission of this manuscript I would like to undertake that the above mentioned manuscript has not been published elsewhere, accepted for publication elsewhere or under editorial review for publication elsewhere; and that all co-authors listed below have agreed to have seen and approved the manuscript for submission.

With regards

Yong Jin Chang, MD

Details of the authors

1. Yong Jin Chang (corresponding author)  
Department of Thoracic and Cardiovascular Surgery, St. Vincent’s Hospital, College of Medicine, The Catholic University of Korea, 93 Jungbu-daero, Paldal-Gu, Suwon, Gyenggi-Do, 442-723, Korea, Tel: +82-31-2498274; fax: +82-31-249; e-mail: tsmate@hanmail.net (Y.J.Chang)

2. Deog Gon Cho  
same as above

3. Kyu Do Cho  
same as above

4. Jae Taek Hong  
Department of Neurosurgery, St. Vincent’s Hospital, College of Medicine, The Catholic University of Korea, 93 Jungbu-daero, Paldal-Gu, Suwon, Gyenggi-Do, 442-723, Korea